Title: A randomized controlled trial of tai chi for long-term low back pain (TAI CHI): Study rationale, design, and methods.

Authors:

Amanda M Hall (amandahall@george.org.au)
Chris G Maher (cmaher@george.org.au)
Jane Latimer (jlatimer@george.org.au)
Manuela L Ferreira (m.ferreira@usyd.edu.au)
Paul Lam (DrPaulLam@taichiproductions.com)

Version: 5 Date: 12 May 2009

Author’s response to reviews: see over
May 12, 2009

Liz Hoffman  
Assistant Editor  
BMC-series journals

Dear Dr. Hoffman,

Re: MS: 1080787280248627

Title: A randomized controlled trial of tai chi for long-term low back pain (TAI CHI): Study rationale, design, and methods.

Thank you for confirming your decision to publish our manuscript pending final revisions. As requested, the following revision has been made in the main text of the most recent version of the manuscript:

Revisions

References - please cite all references in numerical order in the main text. References 34-36 are currently cited out of position.

- This has been corrected on page 7 where refs 34-36 were cited incorrectly, these refs have now been removed from this section. The references for the entire paragraph have been reformatted.

Sincerely,

Amanda Hall